

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Scartascini, Carlos; Stein, Ernesto; Tommasi, Mariano

# Working Paper Political institutions, intertemporal cooperation, and the quality of policies

Working Paper, No. 676

Provided in Cooperation with:

Inter-American Development Bank (IDB), Washington, DC

*Suggested Citation:* Scartascini, Carlos; Stein, Ernesto; Tommasi, Mariano (2009) : Political institutions, intertemporal cooperation, and the quality of policies, Working Paper, No. 676, Inter-American Development Bank, Research Department, Washington, DC

This Version is available at: https://hdl.handle.net/10419/51432

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



INTER-AMERICAN DEVELOPMENT BANK BANCO INTERAMERICANO DE DESARROLLO (BID) RESEARCH DEPARTMENT DEPARTAMENTO DE INVESTIGACIÓN WORKING PAPER #676 RG-N3336

# POLITICAL INSTITUTIONS, INTERTEMPORAL COOPERATION, AND THE QUALITY OF POLICIES

ΒY

Carlos Scartascini\* Ernesto Stein\* Mariano Tommasi\*\*

\* INTER-AMERICAN DEVELOPMENT BANK \*\* UNIVERSIDAD DE SAN ANDRÉS

APRIL 2009

#### Cataloging-in-Publication data provided by the Inter-American Development Bank Felipe Herrera Library

Scartascini, Carlos G., 1971-

Political institutions, intertemporal cooperation, and the quality of policies / by Carlos Scartascini, Ernesto Stein, Mariano Tommasi.

p. cm. (Research Department Working Papers ; 676) Includes bibliographical references. "RG-N3336"

1. Social sciences and state. I. Stein, Ernesto. II. Tommasi, Mariano, 1964-. III. Inter-American Development Bank. Research Dept. IV. Title. V. Series.

HN18 .S438 2009 361.61 S438 -----dc22

©2009 Inter-American Development Bank 1300 New York Avenue, N.W. Washington, DC 20577

The views and interpretations in this document are those of the authors and should not be attributed to the Inter-American Development Bank, or to any individual acting on its behalf.

This paper may be freely reproduced provided credit is given to the Research Department, Inter-American Development Bank.

The Research Department (RES) produces a quarterly newsletter, *IDEA* (*Ideas for Development in the Americas*), as well as working papers and books on diverse economic issues. To obtain a complete list of RES publications, and read or download them please visit our web site at: http://www.iadb.org/res.

#### Abstract<sup>1</sup>

While economists have tended to focus on specific public policies when developing recommendations, the achievement of welfare objectives might depend more on the quality of policies than their content. This paper develops several measures of the qualities of policies across countries, arguing that the quality of public policies depends on each polity's ability to strike intertemporal transactions necessary to develop and sustain effective policies. The analytical framework developed here indicates that this ability depends on several characteristics of political institutions, such as congressional capabilities, judicial independence, and bureaucratic independence and professionalism. The empirical evidence presented supports this idea. The measures of policy quality developed here could be utilized for other purposes, including the determination of conditions under which more public spending in a given area is likely to generate the desired outcomes.

**Keywords:** Political institutions, Public policies, Government capabilities, Intertemporal cooperation, Development, Policy index, Credibility, Judicial independence, Party institutionalization, Congress capabilities, Cabinet stability.

JEL Classification: D72, D78, H10, H50, O10

<sup>&</sup>lt;sup>1</sup> We are particularly grateful to Fabiana Machado for her superb assistance in the development of this document and research agenda, and to discussants and participants at the Public Choice Society Meetings 2009.

## **1. Introduction**

In every state, big or small, new or old, public policies play a fundamental role in influencing economic and social outcomes. In studying the effects of policies on various outcomes of interest, analysts have paid particular attention to the specific content of those policies. Countless overarching prescriptions were generated and also modified over the years according to the conventional wisdom of the time. For example, Latin America has gone through successive policy paradigms in the belief that once the "right" policies are implemented, things will work well. These waves have shifted from State-run inward-looking development in the postwar era to the macroeconomic discipline and trade liberalization of the Washington Consensus of the 1990s. While enthusiasm for the latter has waned, many observers and actors have started to search for (or to prematurely proclaim) a new paradigm.

Some economists' beliefs notwithstanding, a universal set of "right" policies does not necessarily exist. Policies are contingent responses to underlying states of the world. What might work at one point in time in a given country might not work in a different place or in the same place at another time. In some cases, some characteristics of policies (or the details of their implementation) might matter as much as the grand title of the policy. For instance, Dani Rodrik analyzed six countries that implemented a set of policies that shared the same generic title—"export subsidization"—but had widely different degrees of success (Rodrik, 1995). Rodrik relates their success to such features as the consistency with which the policy was implemented, which office was in charge, how the policy was bundled (or not) with other policy objectives, and how predictable the future of the policy was.

The literature on macroeconomic policy offers many such examples, with emphasis on the effects of credibility and flexibility on policy success. The former has been widely recognized in recent work not only on macroeconomics, but also on trade policy, regulation, and other policy areas.<sup>2</sup> The effects of policies on the final economic and social outcomes of interest depends on the actions and reactions of economic and social agents, who take into account their expectations about the future of the policies in question before deciding their responses. As Rodrik explains, in reference to trade reform, "it is not trade liberalization per se, but *credible* trade liberalization that is the source of efficiency benefits. The predictability

<sup>&</sup>lt;sup>2</sup> See, for example, Barro and Gordon (1983), Calvo (1996, Section V), Drazen (2000, Section II), Levy and Spiller (1994), and Rodrik (1989).

of the incentives created by a trade regime, or lack thereof, is generally of much greater importance than the *structure* of these incentives. In other words, a distorted, but *stable* set of incentives does much less damage to economic performance than an uncertain and unstable set of incentives generated by a process of trade reform lacking credibility."<sup>3</sup>

Engerman and Sokoloff (2008) take this argument further by adding to the role of credibility that of policy flexibility in explaining growth: "*Credible* commitment to acknowledge private property rights, whether in the interests of the elite or the majority of the population, is the classic example of the value of certainty about policy action. More generally, however, allowing some *flexibility* in institutions, such that they can be altered to allow private or public agents to take fuller advantage of new opportunities that arise as technology or the environment changes, would be expected to foster improved economic performance and more rapid growth."<sup>4</sup>

The examples above help motivate this project's efforts to build measures of certain characteristics or key features of public policy beyond their specific content (e.g., whether some particular taxes are high or low), which may affect countries' ability to reach desirable outcomes.

Concern with such policy characteristics leads us to focus on the processes that shape policies, carry them forward to implementation, and sustain or adapt them over time. Taking any particular "policy reform" to fruition is a process that involves multiple actors through many stages of the policy process. It requires specific responses from economic and social agents and therefore necessitates several forms of cooperation and positive beliefs about the durability and other properties of the policy. Governments need the ability to formulate and carry out policies, and the capacity to maintain momentum throughout the whole process. Many such "state capabilities" have been referred to in previous literature as key factors in explaining the impact of policies on desired outcomes. In a landmark study, Weaver and Rockman (1993) argue that governmental effectiveness can be measured according to several standards; the one they propose focuses on a set of tasks and on capabilities that governments need, regardless of their specific policy objectives, in order to perform those tasks (Weaver and Rockman, 1993, p. 6). Capabilities are a pattern of government influence on its

<sup>&</sup>lt;sup>3</sup> Rodrik (1989, p. 2). For models formalizing the effects of policies of uncertain duration in several economic contexts, see Calvo (1996, Section V) and Calvo and Drazen (1998).

<sup>&</sup>lt;sup>4</sup> Italics added for emphasis

environment that produces substantially similar outcomes across time and policy areas. Weaver and Rockman propose a number of government capabilities, including *setting and maintaining priorities*, *targeting resources*, *innovating when old policies have failed*, *coordinating conflicting objectives*, *ensuring effective implementation*, and *ensuring policy stability* so that policies have time to work.

In this paper we propose an explanation of such capabilities based on a framework of intertemporal cooperation. The framework (developed in Section 2) indicates that policies with good characteristics such as stability, adaptability and coherence are likely to emerge when policymaking actors can make and sustain agreements over time. In turn, the capacity to achieve such cooperative agreements (cooperative equilibria to repeated policymaking games) depends on some characteristics of the institutions of policymaking, including the policymaking capabilities of congress, judicial independence, party system institutionalization, and the like.

In the next section, we present the framework that guides both the construction of the variables and the empirical exercises. Section 3 introduces a number of variables that capture various qualities of public policies. Section 4 introduces a number of variables that, according to our framework, capture aspects of the workings of institutions relevant to explain the quality of policies. Section 5 presents the empirical results, which show that institutional variables likely to induce intertemporal cooperation matter for explaining good policy outcomes. Section 6 concludes and suggests various applications of the measures of policy quality developed in the paper.

#### 2. The Framework<sup>5</sup>

The process of policymaking in modern-day democracies can be understood as a process of bargains and exchanges (or transactions) among various political actors. Some of these exchanges are consummated instantly (spot transactions), while in many other cases current actions or resources are exchanged for promises of future actions or resources (intertemporal transactions). Issues of credibility and the capacity to enforce political and policy agreements are crucial for political actors to be able to engage in intertemporal transactions.

<sup>&</sup>lt;sup>5</sup> This framework is presented in more detail in Spiller, Stein and Tommasi (2003) and Chapter 2 of Spiller and Tommasi (2007), building upon previous contributions such as Alesina (1988), Dixit, Grossman and Gul (2000), Dixit (2003), and de Figueiredo (2002).

The framework suggested here is an elaboration of previous work on transaction cost economics and its application to politics.<sup>6</sup> A number of features, amenable to analysis from a transaction cost perspective, characterize the political transactions surrounding public policies. We mention six that are captured in a stylized manner by our framework:

- (1) Politics and policymaking take place over time. Decisions are made at different points in time, often by different configuration of actors, and decisions made at any point in time have both short-term and long-term consequences.
- (2) The relative political power of various actors changes over time.
- (3) There are elements of both conflict and commonality of interests in almost any relevant policy issue.
- (4) The socioeconomic reality on which policies operate changes over time. Random events require policy adaptation. New circumstances in international markets, policy decisions in other countries, technological changes, diseases, natural disasters, and social and demographic changes usually require new policies or adjustment of previous ones.
- (5) Most policies could be characterized by two decision frequencies: moments of major institutional definition ("contractual moments") and regular policymaking under those rules.
- (6) Many of the changing realities in (4) are such that it would be impossible for political or policy agreements to cover every feasible future circumstance. Thus political contracts (5) are necessarily incomplete.

In order to capture those features, we depict policymaking as the outcome of a repeated game (a simple example of which is formalized in the Appendix).<sup>7</sup> Imagine a number of political actors who have to make some collective decision. These players have a common interest in having the policy respond to an *economic or technological shock*. At the same time, the heterogeneity of preferences and/or distributive nature of politics generate

<sup>&</sup>lt;sup>6</sup> North (1990) and Dixit (1996) have labeled *transaction cost* politics the use of transaction-cost reasoning to think about politics. Related work in political science has been pioneered by Weingast and Marshall (1998), Moe (1990), Epstein and O'Halloran (1999), and Huber and Shipan (2002).

<sup>&</sup>lt;sup>7</sup> The model is more fully developed in Spiller and Tommasi (2007, Chapter 2).

conflict. The relative power of different players in the collective decision process changes over time, according to a *political random shock*. (Imagine some variation of a random recognition rule *a la* Baron and Ferejohn, 1989).

Assume that the political game starts with an initial period, before the play of the repeated game, in which players can make some agreements by unanimity. This captures the notion of a "contracting moment," a time when the parties possibly agree on some restrictions on the future play of the game. The set of feasible contracts at that moment will depend on a number of things, including the availability of enforcement technologies—for instance, whether there is an independent judiciary.

Define first best policies as those that would be agreed upon in a complete contract before the world starts running—or equivalently, those that a benevolent social planner would choose. It is easy to show (see Appendix) that these optimal policies will be "moderate" and invariant to the realization of political shocks,<sup>8</sup> but flexible enough to adjust to economic shocks. It is also easy to see that if political actors are durable and patient enough, they can sustain first-best policies as a Nash equilibrium in an infinitely repeated game.<sup>9</sup>

On the other hand, if their discount rate is high enough, (full) cooperation will not be sustainable in equilibrium. In non-cooperative equilibria, such as the infinite repetition of the unique Nash equilibrium of the one-shot game, each party that has political power at a given point in time will maximize its own welfare without any intertermporal considerations for those holding power in the past or in the future. In such a case, policies will depend on the realization of political shocks, and welfare will be lower than in the cooperative case.

Looking into the prior contracting stage in which players can make some agreements, restrictions on the set of feasible (i.e., enforceable) contracts will depend on the issues in question and on available enforcement mechanisms. Suppose, for instance, that agreements can be enforced by third parties, but that the realization of economic shocks is not verifiable. In that case, it will not be possible to enforce agreements that prescribe (economic) state-contingent rules. Simple rules, however, can be agreed upon. These rules would imply relatively inflexible policies. Since *ex-ante* parties prefer policies that are independent of

<sup>&</sup>lt;sup>8</sup> That is, first best polices will not depend so much on the identity of the each period's agenda setter in Baron and Ferejohn (1989) or of the party in power in Alesina (1988). This is the *policy* equivalent of results on the stability of the *polity* in works such as Przeworski (2005) or Wantchekon (2000).

<sup>&</sup>lt;sup>9</sup> As discussed later, the possibility of sustaining cooperation will depend on a number of factors beyond the discount rate.

political shocks, these simple rules will not be sensitive to those shocks (which is good). On the other hand, since economic shocks are not verifiable, policies will not be able to adjust to the changing economic environment either (which is bad). The best *ex-ante* policies, then, may be rigid policies. As a consequence, they deliver lower welfare levels than could be obtained in a fully cooperative equilibrium or in an environment of complete enforcement.

Indeed, whenever the repeated game delivers full cooperation, these simple rigid rules will not be utilized—players will prefer adaptation to economic shocks to a set of rigid *exante* rules. When the repeated game does not develop cooperation, though, there are conditions under which an inflexible policy rule will be chosen over the discretionary policy of the Nash equilibrium. That choice depends on the relative cost of not being able to adjust policies to economic shocks (related to the variance of the economic shocks), compared to the cost of "partisan" policymaking (related to the heterogeneity of preferences).<sup>10</sup> Thus, when enforcement of intertemporal political exchanges is relatively weak, we may observe highly volatile political agreements or highly inflexible policies. (As argued in Spiller and Tommasi 2007, Argentina is a clear example of a country with non-cooperative policymaking leading to volatile policies occasionally curtailed by very rigid credibility-enhancing mechanisms, such as the monetary Convertibility regime).

We can obtain a further connection between the degree of cooperation in equilibrium and various features of public policy by expanding the model in a couple of directions. Two natural extensions consist of: (i) adding intertemporal policy linkages, and (ii) introducing individual policy actions (by different layers of government in a federal hierarchy, by different horizontal units such as ministries in a given level, by multiple actors throughout the policy process, etc.).

Many policies are linked over time (i). Those linkages could arise because of technical reasons (i.e., policies that have intertemporal effects), legal reasons (a law is in place until it is changed), or economic reasons (present fiscal actions have future effects through intertemporal budget constraints). Introducing such linkages, Spiller and Tommasi (2007) show that in bad transaction environments, some welfare-improving policies (or policy

<sup>&</sup>lt;sup>10</sup> See the Appendix. Notice that this result is similar to the standard rules vs. discretion result in monetary policy. See for instance Persson and Tabellini (2000), Chapter 17.

reforms) are not undertaken, and that there is under-investment in policymaking capacities.<sup>11</sup> The former result obtains due to an inability to instrument the intertemporal compensations necessary to improve the welfare of all veto players. The latter is just the "policy" analogue of the well-known result in transaction cost economics that *ex-post* opportunism reduces *ex-ante* investment.

Another easy extension of the model (ii) introduces individual policy actions other than participation in the "collective" choice discussed above. Imagine, for instance, the multiple policy authorities in a federal system, or the various ministries in charge of the implementation of different policies (and the constituencies or iron triangles behind them). Those actions could be more or less cooperative. In bad transaction environments, those individual policy actions will be less cooperative (a basic result from non-cooperative game theory), leading to poorly coordinated (or "balkanized") policies.

To summarize, we have argued that in less-cooperative policymaking environments policies might be too volatile and/or too rigid, poorly coordinated, and in general of low quality due to insufficient investment. These properties of policies are among the dependent variables we explore in Section 3. After doing that, in Section 4 we come back to discuss, on the basis of the framework just presented, the type of institutional variables that will constitute our explanatory variables.

## 3. Characteristics of Public Policies

The framework just presented generates specific predictions about some policy characteristics. That logic has guided our data generating efforts. Since most policies are not one-shot decisions independent of past and future policy decisions, policy characteristics will depend on the ability of the polities to reach certain intertemporal agreements which would allow making decisions today that are consistent over time. As in transaction cost economics poor transaction environments lead to less investment than better ones, in our framework poor transaction environments would favor policies that are too volatile (they will change too often in response to political winds); too rigid (often incapable of adjusting in the face of changed circumstances, such as economic shocks); and poorly coordinated or incoherent. Also, polities

<sup>&</sup>lt;sup>11</sup> By investment in policy-making capacities we refer to things such as developing a competent bureaucracy, or legislators investing in acquiring policy expertise. This is consistent with results in Besley and Persson (2007).

afflicted by bad transaction environments will invest insufficiently in capacities and thereby produce low-quality policies.

Following that logic we have selected the following six indicators of a country's policy characteristics: stability, adaptability, coherence and coordination, implementation and enforcement, efficiency, and public-regardedness.

**Policy Stability**: Some countries seem capable of sustaining most policies over time. In other countries, policies are frequently reversed, often at each minor change of political winds (whether a change in administration or a change in some key cabinet member or senior bureaucrat). Having stable policies does not mean that policies cannot change at all, but rather that changes tend to respond to changing economic conditions or to failure of previous policies, rather than to flimsy political winds. In countries with stable policies, changes tend to be incremental, building upon achievements of previous administrations, and tend to be achieved through consensus. In contrast, volatile policy environments are characterized by large swings and by lack of consultation with different groups in society.

**Policy Adaptability:** It is desirable for countries to be able to adapt policies to changing economic conditions and to change policies when they are clearly failing. Policy adaptability can be hindered either by a policy making process prone to gridlock, or by rigidities introduced explicitly to avoid opportunistic manipulation of policy. In some polities, the configuration of the political system can lead to gridlock, making it difficult to achieve change. In other cases, the government of the day might be prone to abuse discretion by adopting opportunistic one-sided policies. In order to limit that opportunism, such polities might resort to fixed policy rules that are difficult to change.<sup>12</sup> This, of course, would limit policy volatility, but at the cost of reducing adaptability. Generated by either gridlock or built-in rigidities, low policy adaptability leads to the inability to respond to shocks adequately, and a propensity to keep sub-optimal policies for extended periods of time.

**Policy Coordination and Coherence:** Public policies are the outcome of actions taken by multiple actors in the policymaking process. Ideally, different agents acting in the same policy domain should coordinate their actions to produce coherent policies. However,

<sup>&</sup>lt;sup>12</sup> This is sometimes accomplished by embedding policies (such as pension benefits in Brazil or intergovernmental transfers in Argentina) into the constitution.

this does not always occur. In some countries, policymaking on certain issues involves a large number of actors that do not communicate adequately with each other, leading to what Cox and McCubbins (2001) have called "balkanization" of public policies. Lack of coordination often reflects the non-cooperative nature of political interactions. It may occur among different agencies within the central government, between agencies in the central government and others at the regional or municipal level, or even among agents that operate in different stages of the policymaking process—different interest groups with differential access to the Executive, Legislature, bureaucracy, or Judiciary, will exercise their pressure at the stage which is most favorable to them.

**Policy Implementation and Enforcement:** A policy could be well thought out and pass through the appropriate legislative debate, and yet be completely ineffective if it is not well implemented and enforced. In many countries, the quality of policy implementation and enforcement is quite poor. This is associated in part with the lack of capable and independent bureaucracies, as well as the lack of strong judiciaries. To an important degree, the quality of policy implementation and enforcement in a given country will depend on the extent to which policymakers in that country have incentives and resources to invest in such policy capabilities.

**Policy Efficiency:** Whatever policy direction a government decides to follow (redistribute to the poor, clean the environment, promote non-traditional exports), it can do so with varying degrees of efficiency—that is, by making better or worse use of its human and economic resources. Efficient policies imply, for example, that public spending is not wasteful. Efficient policies might not necessarily be public-regarded (the government could be very efficient in targeting a very small subset of the population).

**Public-Regardedness of Policies:** Public-regardedness refers to the extent to which policies produced by a given system promote the general welfare and resemble public goods (that is, are "public-regarding") or tend to funnel private benefits to certain individuals, factions, or regions (Cox and McCubbins, 2001).

# 4. Political Institutions and Intertemporal Cooperation

We argued in Section 2 that the ability of political actors to cooperate will be an important determinant of the characteristics of policies in each country—characteristics such as those presented in Section 3. The next question, then, is what conditions make policy cooperation more likely.

It has already been established that the likelihood of political cooperation depends on the patience of the players (i.e., on their discount factor). The repeated-game approach sketched above and exemplified in the Appendix could be utilized to investigate various other elements of the description of the game that could facilitate (or hinder) the enforcement of cooperative play. We list here some factors that affect the degree of cooperation in equilibrium outcomes, drawing insights from the theory of repeated games.<sup>13</sup>

## Number of Political Players

The larger the number of players, the smaller the set of other parameters for which cooperation obtains. Fudenberg and Tirole (1991: Section 5.1.2) and Fudenberg and Maskin (1986) show that, holding constant the set of feasible payoffs, increasing the number of players reduces the set of equilibria towards less cooperative ones.

#### Intertemporal Linkages among Key Political Actors

The intertemporal pattern of interactions among specific individuals in formal political positions (such as legislators, governors, and bureaucrats) matters for developing cooperative outcomes. It is not the same to have a legislature in which the same individuals interact over extended periods of time as to have a legislature where individuals are drawn at random from given populations (parties, provinces, etc.) with frequent replacement. Cooperation is less likely in the latter. (Also, historical events, such as past democratic history can leave a legacy of short-termism, as argued in Dominguez and Lowenthal, 1996.)

#### Delegation

Other than self-enforcement through repeated play, certain forms of cooperation could be achieved by alternative institutional means. Delegating policy to an independent bureaucracy

<sup>&</sup>lt;sup>13</sup> See for instance, Green and Porter (1984), Rotemberg and Saloner (1986), Fudenberg and Tirole (1991: Chapter 5), and Mailath and Samuelson (2006). Dixit (1996: 71) presents a similar listing.

is one such alternative. In the example of the Appendix, it is easy to show that delegating policy forever to an individual with preferences falling in between those of the two parties leads to the first best. More generally, delegation has its problems, but there will be cases in which the cost of those problems is smaller than the cost of "partisan" policymaking.

#### Availability of Enforcement Technologies

As in transaction cost economics, intertemporal cooperation is easier to achieve if there is good third-party enforcement. The presence and characteristics of a potentially impartial umpire and enforcer of political agreements, such as an independent judiciary, will vary from country to country, providing variance in the degree of enforcement of intertemporal political cooperation.

#### Characteristics of the Arenas Where Key Political Actors Undertake Their Exchanges

The complex intertemporal exchanges required for the implementation of effective public policies could be facilitated by the existence of exchange arenas that are organized in ways that make cooperation easier to enforce. Seminal work on the U.S. Congress debates the role that different institutional arrangements have in facilitating legislative bargaining, but it is agreed that *somehow* things are arranged in a way that facilitates some intertemporal cooperation in political exchanges (see for instance Weingast and Marshall, 1988; Shepsle and Bonchek, 1997; and the collection in Shepsle and Weingast, 1995). Whether the legislature as the arena where these transactions take place is adequately institutionalized depends on several factors, including legislators' incentives and capabilities.

To sum up, political cooperation leading to effective public policies is more likely if: (1) the number of political actors is small, (2) those actors have long horizons and/or strong intertemporal linkages, (3) good delegation technologies are available, (4) good enforcement technologies (such as a strong court to arbitrate) are available, and (5) the key political exchanges take place in arenas where properties (1)-(4) tend to be satisfied.

What are the actual political institutions or characteristics of political institutions that make political cooperation (and hence good public policies) more or less likely? The theoretical elements of the description of the game listed above could be mapped to observable features of the policymaking environment. Here we concentrate on a few such features which seem to capture some of the determinants of political cooperation and (as shown in the next section) for which there are proxy variables available in international datasets with wide enough country coverage. These characteristics are the following:

**Congressional Policymaking Capabilities:** Congress is the democratic arena par excellence for the bargaining and (hopefully) enforcing of intertemporal policy agreements. Legislatures are critical to the functioning of democracy and act as an important arena for discussing and negotiating policy. A legislature made of up professional legislators (with technical capabilities for discussing and overseeing policies) and adequate organizational structures can facilitate the development of relatively consensual and consistent policies over time. From applied analysis of Latin American policymaking (reflected in Stein et al., 2008 and in Saiegh, 2009) we have identified a number of organizational characteristics of legislatures in the policymaking process. Those measures, in turn, correlate well with some variables available for a large sample of countries, capturing opinion about the effectiveness of lawmaking bodies and confidence in Parliament, variables that we use in the empirical implementation of the next section.

**Judicial Independence:** As stated above, the Judiciary, especially the Supreme Court or Constitutional Tribunal is a natural candidate for the enforcement of those political or policy agreements reflected in constitutions and laws. In its role as an independent referee, the judiciary can provide a "durability mechanism" that can increase the probability of reaching intertemporal agreements.<sup>14</sup> A judiciary that effectively plays its role may contribute to better public policy outcomes, such as enhanced policy stability, and policy implementation and enforcement. If the judiciary is not independent of the other branches of government, it may not be effective in fulfilling that role.

**Civil Service Capacity:** A strong, independent and professional bureaucracy seems the most natural vehicle for the flexible enforcement of political agreements via delegation. An effective and capable bureaucracy is likely to improve the quality of implementation of public policies, as well as their coordination across ministries. The competence and independence of the bureaucracy may decrease the likelihood that policy will be prone to

<sup>&</sup>lt;sup>14</sup> This view follows the work first discussed by Landes and Posner (1975) in which the presence of an independent court tends to resolve time-inconsistency problems (that is, agreements made today have a higher discounted value because they are less likely to be changed in the future). See also Crain and Tollison (1979) and Crain (2001). The role of intertemporal *political* enforcement is explored, in the context of Latin American judiciaries, in Souza (2009).

politicization and political opportunism, and could increase policy adaptability to changing circumstances by relying on technical expertise (Zuvanic and Iacoviello, 2009).

**Party System Institutionalization:** An institutionalized party system is a natural aggregator that reduces the effective number of players at the bargaining table and increases the horizons of individual political actors. The structure and organization of political parties and party systems can have an important influence on the policymaking process, both by playing a direct role and through interactions with other institutions. Political parties can influence policy debates, affect executive-legislative relations, enhance or constrict the possibilities for coordination in congress, or manage the incentives of politicians at both the national and local level. In sum, institutionalized party systems serve as facilitators of intertemporal policy compromise.<sup>15</sup>

**Cabinet Stability:** Cabinet ministers in many countries play key roles in various stages of the policy process. The strength and organizational abilities of cabinets can have important effects on the outcomes of public policy. For example, a certain degree of cabinet stability is likely to be necessary to promote longer-term policies and allow ministers to see programs and policy implementation through to completion. Frequent turnover of cabinet ministers may foster the short-term orientation of policy and frequent policy changes, as well as a reduction in the effective coordination between the ministers and the bureaucratic institutions they may oversee (Martínez-Gallardo, 2009).<sup>16</sup>

## **5.** Empirical Analysis

#### 5.1 Policy Characteristics across Countries

In this paper we attempt to build cross-country indicators of policy capabilities drawing from available broad cross-national sources. The construction of this international dataset builds upon a previous and more in-depth dataset (presented in Inter-American Development Bank, 2005) in which these policy characteristics were constructed for 18 Latin American countries combining international sources and our own survey of experts, double-checking with specific

<sup>&</sup>lt;sup>15</sup> In previous work within Latin America we have found that institutionalized party systems, if programmatic, tend to correlate with high-quality policies (Stein and Tommasi, 2007). See also Jones (2009), Mainwaring and Torcal (2005), and Mainwaring and Scully (1995).

<sup>&</sup>lt;sup>16</sup> There are other variables, such as the legislative reelection rate, that proxy very well some of the features of cooperation (for example, it is probably the best measure for approaching the horizon of legislators and the existence of intertemporal linkages) and we would have liked to include in the analysis. Unfortunately, there is no widely available cross-country data on legislative reelection rates.

studies of various countries (Stein et al., 2008) and policy areas (taxation, regulation, education, social policy). The dataset we present in this paper is an extension of the international components of the previous Latin American dataset. The indexes presented here have a high level of correlation with the other components of the Latin American indexes for those 18 countries. Most of the information presented here comes from qualitative surveys which we have complemented and double-checked with more objective data whenever possible.<sup>17</sup> For brevity, we don't describe the data sources in this document. For complete references, links and documentation see Berkman et al. (2009).

The policy characteristics were constructed as follows:

*Policy Stability*: To gauge policy stability we used four variables from three different sources. The first is the standard deviation of the detrended Fraser Index of Economic Freedom for the years 1999 to 2004. Two of the variables come from the Global Competitiveness Report (GCR) of 2002. One measures whether legal or political changes over the past five years have undermined respondent firms' planning capacity, and the other measures whether new governments honor the contractual commitments and obligations of their predecessors. Finally we used a question from Profils Institutionnels (PI) where experts evaluate the "Consistency and continuity of government action in economic matters."

*Policy Adaptability*: Our measure of policy adaptability was constructed based on four variables from three different sources. Two variables come from the Columbia University State Capacity Survey (CUSCS). In the first question, experts (from academia, government and media) rate states' ability to respond effectively to domestic economic problems, and in the second they rate states' ability to formulate and implement national policy initiatives. A third variable is drawn from the Bertelsmann Transformation Index (BTI) for 2006 measuring the degree of adaptability based on the ability of the political leadership to act flexibly, political leaders' capability for learning, and whether political leaders can replace failing measures with innovative policy. Finally, we used the Profils Institutionnels item where

<sup>&</sup>lt;sup>17</sup> For instance, we have cross-checked the correlation of our measures of some of these policy characteristics with indicators based on more objective (say, fiscal) data, which are in some cases available for (much) smaller samples of countries. (See the next footnote for some examples). In order to verify that the data we collected were reliable we ran a number of checks to examine whether we were measuring the desired components of public policies and institutions. These exercises are reported in the working paper Scartascini, Stein, and Tommasi (2008b).

experts evaluate the decision-making capacity of political authorities in economic matters (responsibility, rapidity, etc).<sup>18</sup>

*Policy Coordination and Coherence:* Our measure of coordination and coherence was built based on two variables, one from the Columbia University State Capacity Survey and the other from the Profils Institutionnels database. The first is a rating of the effectiveness of coordination between the central government and local-level government organizations. The second rates co-ordination between ministries and within administrations.

*Policy Implementation and Enforcement:* This index is based on the following six variables. Expert evaluation of whether the minimum wage set by law in the country is enforced, expert evaluation of whether tax evasion in the country is rampant or minimal, and expert evaluation of whether environmental regulation in the country is enforced, all from the GCR. We draw from the BTI analysts' estimate of whether the government implements its reform policy effectively, and we draw from the CUSCS a rating of states' ability to formulate and implement national policy initiatives and a rating of states' effectiveness in collecting taxes or other forms of government revenue.

*Policy Efficiency*: To capture efficiency we employed three measures. From the GCR we used the expert ratings of the composition of Public Spending and whether it is wasteful. From the BTI we employed experts' evaluation of whether the government makes efficient use of available economic and human resources. Finally, from the Economist Intelligence Unit (EIU) we used the experts' assessment of the effectiveness of the political system in formulating and executing policy.

*Public-Regardedness of Policies:* The public-regardedness index is captured by three variables. The GCR's expert rating of whether when deciding upon policies and contracts, government officials usually favor well-connected firms and individuals or are neutral among firms and individuals. Also from GCR we use the experts' evaluation of whether government social transfers go primarily to poor people or to the rich. Finally, we include the Transparency International Corruption Perceptions Index (2005).

<sup>&</sup>lt;sup>18</sup> One of the double checks we did was to verify that our index of adaptability correlates well with indicators of whether social policy adjusts well over the macroeconomic cycle, for those countries for which we had such measures; see Braun and Di Gresia (2002). Similar checks with fiscal variables and variables related to education policy were performed in Diaz et al. (2009).

*The Policy Index*: The various indexes we have constructed to measure these key features could be combined in different ways to come up with an overall index of the quality of public policies, which should provide a good picture of the quality of policymaking in many countries. Because we do not have a prior regarding which index should be weighted most heavily, in constructing a policy index for this study we allocated the same weight to each of the features discussed above and use the simple average of the different policy characteristics.<sup>19</sup>

All the variables we constructed are available in an accompanying dataset (Berkman et al., 2009). Most of the variables are positively correlated, suggesting that to a great extent all good things tend to go together. In Scartascini, Stein, and Tommasi (2008b) we argue that each one of them measures a substantively different concept and different policy dimensions are indeed being captured by the measures introduced here.<sup>20</sup> We also show in that working paper that these policy features are associated with relevant development outcomes such as GDP per capita growth and change in the UNDP Human Development Index.

As indicated in Table 1, countries tend to rank as expected in these policy characteristics (e.g., the developed on top and Sub-Saharan Africa at the bottom). The developed countries (including Western Europe, Japan, Australia, the United States, and Canada) rank the highest in terms of the policy index and are considerably higher on average than those of the rest of the world. The second highest ranking is that of the countries of East Asia and the Pacific, which includes the economically successful countries of Singapore, Taiwan, Thailand, and China. Following third and fourth are those of the Middle East/North Africa and Eastern Europe/Central Asia, while Latin American and Caribbean (LAC) countries fall to fifth on the scale. The only regions surpassed by LAC are Sub-Saharan Africa and South Asia. The fact that the indicators of policy quality are related to the level of development is controlled for in the empirical analysis of Section 5.3.

<sup>&</sup>lt;sup>19</sup> We have explored various weighting alternatives, even using a policy index including only on the first four of these policy characteristics, which come out more directly from the framework of Section 2. The empirical results are roughly the same for all such exercises.

<sup>&</sup>lt;sup>20</sup> We have taken great care in ensuring that what we have is not just an optimism/pessimism bias driven by third factors. One of the reasons for our confidence is that the different components of the indexes come from different sources and have been averaged (when possible) over long periods of time.

Region	Stability	Adaptability	Implementation	Coordination	Efficiency	Public Regardedness	Policy Index
Developed Countries	3.39	2.80	3.01	2.88	2.44	2.97	2.96
East Asia/Pacific	2.74	1.79	2.10	1.84	1.81	1.61	1.97
Middle East/North Africa	2.62	1.37	1.93	1.69	1.55	1.79	1.82
Easter Europe/Central Asia	2.65	1.64	1.86	1.53	1.46	1.39	1.7
Latin America/ Caribbean	2.52	1.56	1.77	1.26	1.29	1.35	1.62
South Asia	2.83	1.44	1.49	1.19	1.32	1.16	1.53
Sub-Saharan Africa	2.58	1.21	1.76	1.11	1.29	1.32	1.53

 Table 1. Average Policy Characteristics per Region

Note: Regions ordered according to their standing in the Policy Index (which takes values between 1 and 4).

This distribution is similar when we look at individual policy features. However, changes in the rankings seem to suggest that even though all the variables tend to go together, the polities of some countries seem to be better able to deliver certain features of policies to the detriment of others. Indeed, as shown in Table 2, while in some features countries of a given region tend to be close to each other, in others the variance is relatively large. To give some examples, developed countries tend to receive relatively high scores on stability. If we run cluster analysis on the data we find that they tend to form a somewhat uniform group that stands out from countries in other regions (the variance is 0.03). However, if we apply the same procedure to coordination the picture is quite different (variance is 0.48). Countries like Italy, Israel and France tend to be closer to the success cases in the developing world (e.g. Botswana, Chile, Brazil, and Taiwan) rather than to the highest-ranking developed countries (e.g. Finland, Germany, Sweden, and Norway). Conversely, Singapore tends to cluster together with successful developed countries, rather than with its East Asian fellows, in features such as coordination and implementation and enforcement.

Public Stability Adaptability Implementation Coordination Efficiency Regardedness Policy Index Reaion **Developed Countries** 0.03 0.21 0.2 0.48 0.14 0.2 0.13 East Asia / Pacific 0.1 0.4 0.32 0.51 0.48 0.27 0.63 Easter Europe / Central Asia 0.1 0.35 0.19 0.28 0.17 0.1 0.15 \_atin America/ Caribbean 0.17 0.29 0.26 0.32 0.25 0.23 0.33 Middle East/ North Africa 0.13 0.13

0.1

0.26

0.36

0.16

0.41

0.2

0.05

0.27

0.1

0.11

0.18

0.05

0.18

0.08

0.07

0.07

South Asia

Sub-Saharan Africa

0.25

0.14

0.32

Table 2. Variance of Distances among Countries within Regions<sup>21</sup>

<sup>&</sup>lt;sup>21</sup> These variances are calculated based on the mean of Euclidean distances between each country of a given group and each of the other countries in that same group.

Again, these differences in clustering show that some countries seem to be better able to deliver certain features of policies to the detriment of others. For example, countries like Korea seem to have a high capacity to adapt their economic policies; however, they seem to be less able to do so in line with benefits for the overall population (public-regardedness). Comparatively, Finland seems to favor a wide range of the population with its policies; however, its capacity to adapt in the face of shocks seems to be relatively lower.

One of the concerns usually expressed by researchers is that, despite the multitude of measures of policy and institutional capacity available, they are all getting at the same abstract concept (Van de Walle, 2005; Knack and Manning, 2000). In this regard, the patterns discussed above seem to contribute to the validity of the measures proposed here to the extent that interesting and reasonable variations are observed when comparing country rankings on the different dimensions.

#### 5.2 The Workings of Political Institutions

As we have argued above, the ability of countries to achieve good policy characteristics depends on the quality of its institutional environment and the ability of the actors in charge of policymaking to reach intertemporal agreements. In this paper, we proxy the conditions for cooperation with some characteristics of key players and arenas such as congress, the party system, the judiciary, and the bureaucracy, following the logic of Section 4.

**Congressional Policymaking Capabilities:** We used the average of two data sources: the effectiveness of lawmaking bodies (from the GCR) and the population's confidence in parliament (from the World Values Survey).<sup>22</sup>

**Judicial Independence:** This variable has been constructed from three different sources—GCR, BTI, and the Fraser Index—that attempt to measure the same phenomenon: whether the judiciary is subject to interference by the government or other political actors.

**Civil Service Capacity:** This variable has been constructed using data from two sources—the State Capacity Survey and the International Country Risk Guide—and includes indexes that measure the degree of professionalism in the civil service, whether recruitment is

 $<sup>^{22}</sup>$  For this variable, as for most of the others, we have found a strong correlation within the Latin American subsample between these measures and more focused measures constructed for those 18 countries. Saiegh (2009) and Inter-American Development Bank (2005) show that these two measures correlate well with indicators of the duration of legislative careers, legislators' education, and the degree of specialization in congressional committees.

based on merit, the level of the bureaucracy's functional capacity and performance, and its efficiency.

**Party System Institutionalization:** The Party System Institutionalization Index is comprised of five variables, which measure the extent to which there is a stable, moderate and socially rooted party system that can articulate and aggregate societal interests (from the BTI); the level of confidence in political parties (from the World Values Survey and various Barometers); vote volatility; the age of parties; and the fairness of elections.

**Cabinet Stability:** Defined as the number of times in a year that a new premier is named and/or 50 percent of the cabinet posts are occupied by new ministers,<sup>23</sup> from the Cross National Time Series database.

**Other Institutional Variables:** The variables listed above are natural proxies for some facilitators of intertemporal cooperation. In our analysis we have included some of the institutional rules used more broadly in the literature on political institutions and policy, such as the political regime, the electoral system, and the effective number of parties in the legislature. These variables, which are widely used in the Political Economy literature (e.g., Persson and Tabellini, 2003) are not easily mapped directly into this project's motivating framework, and as will be shown below, they are usually not related to the policy variables.<sup>24</sup>

#### 5.3 Relating Political Institutions and Policy Outcomes

The framework used to construct these variables generates a number of predictions relating some institutional conditions likely to foster intertemporal cooperation to the features of policies captured by our policy indexes. As shown in Table 3 (each cell in the table shows the coefficient for the variable of interest taking the institutional variables one at a time and controlling for GDP per capita and regional dummies), our expectations are borne out by the data.<sup>25</sup> The "intertemporal" institutional variables are often positively and significantly

<sup>&</sup>lt;sup>23</sup> We converted the original variable so higher numbers indicate higher stability.

<sup>&</sup>lt;sup>24</sup> Scartascini (2007) develops the potential links between the institutional variables traditionally utilized in the literature and the features of cooperation. Gerring, Thacker, and Moreno (2005), another interesting inquiry into the role of alternative political institutions on some governance outcomes, shares the same broad set of concerns as this agenda. The authors propose a reduced-form characteristic of political systems called *centripetalism* as the most favorable to good governance. In their view, parliamentarism favors centripetalism, and hence good outcomes.

<sup>&</sup>lt;sup>25</sup> Those countries defined as non democracies since 1990 are excluded from the empirical exercises. Democratic data are taken from Freedom House.

correlated with policy features. This suggests these desirable policy features might indeed be a consequence of good well-functioning policymaking institutions being in place.<sup>26</sup>

		icacionai	, al lable a				
Policy Characteristic	Stability	Adaptability	Implementation	Coordination	Efficiency	Public	Policy Index
Institutional Variable			-			Regardedness	-
Congressional Policy making Conchilition	0 420***	0 465***	0 479***	0 420***	0 570***	0 509***	0 51 4***
Congressional Policymaking Capabilities	0.420	0.465	0.4/0	0.439	0.570	0.506	0.014
Observations	(0.0948)	(0.141)	(0.117)	(0.149)	(0.111)	(0.111)	(0.0830)
Ubservations	84	87	91	80	92	92	92
K-squared	0.671	0.613	0.693	0.704	0.652	0.802	0.826
Party System Institutionalization	0.142	0.645***	0.431***	0.364**	0.184	0.257**	0.337***
	(0.120)	(0.150)	(0.131)	(0.175)	(0.137)	(0.125)	(0.102)
Observations	89	92	96	84	96	95	97
R-squared	0.579	0.652	0.670	0.682	0.553	0.766	0.778
Judicial Independence	0.329***	0.361***	0.468***	0.384***	0.324***	0.497***	0.426***
	(0.0600)	(0.0951)	(0.0697)	(0.102)	(0.0753)	(0.0660)	(0.0493)
Observations	88	91	95	83	96	95	96
R-squared	0.692	0.636	0.755	0.727	0.623	0.851	0.865
Civil Service Capacity	0.245***	0.478***	0.464***	0.499***	0.257***	0.294***	0.388***
	(0.0674)	(0.0878)	(0.0693)	(0.0920)	(0.0802)	(0.0807)	(0.0536)
Observations	88	90	94	84	93	93	94
R-squared	0.635	0.687	0.757	0.758	0.594	0.787	0.845
Cabinet Stability	0 163	0 380	0 417*	0.812***	0.458*	0 595***	0 446***
Cabinet Clability	(0.185)	(0.260)	(0.222)	(0.257)	(0.233)	(0.214)	(0 170)
Observations	(0.100)	(0.200)	(0.222)	(0.207)	(0.200)	(0.214)	(0.170)
Diservations Required	0.576	92	90	0 702	90	95	0.760
K-Squared	0.576	0.566	0.044	0.703	0.562	0.775	0.769

 Table 3. Institutional Strengths and the Characteristics of Policies,

 One Institutional Variable at a Time

Standard errors in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Controls included in the regressions are: Ln(GDPpc) in 1990 and 7 regional dummies

Similar results are obtained when we look at the multivariate interaction of these variables. We focus this brief analysis on the Policy Index.<sup>27</sup> Looking at the specifications in Table 4, where each includes a particular set of variables, we see that most of the institutional strength variables have a significant effect on the overall quality of policies. The exceptions are the measures of party institutionalization and cabinet stability, where we fail to reject the hypothesis that their effect is null for the policy index. They are significant, however, for some of the individual indexes. Plain institutional rules have no discernible effect either, and only the previously significant institutional quality variables remain positive and significantly associated with the Policy Index. For this analysis we use three variables:<sup>28</sup> Political Regime (whether the system is presidential or parliamentary), Proportional Electoral System, and the

 $<sup>^{26}</sup>$  As shown below, when we look at basic institutional rules such as the electoral system or the government system we fail to observe significant effects (even if the institutional characteristics we have introduced are not included in the regressions).

<sup>&</sup>lt;sup>27</sup> See Machado et al. (2009) for a more complete analysis, looking at each of the dependent variables in detail, contrasting the predictions of this framework with those of other approaches that speak to each of those issues.

<sup>&</sup>lt;sup>28</sup> We have also found the same results with other variables such as bicameralism, and federalism, which we do not report here.

Effective Number of Parties, all of which come from the Database of Political Institutions.<sup>29</sup> Finally, we also include some economic controls related to exposure to economic volatility and shocks. Including these variables does not change the results.

Dependent Variable	Policy Index							
Independent Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Congressional Policymaking Capabilities	0.175*	0.170*	0.155	0.222**	0.186*	0.174*	0.175*	
	(0.0883)	(0.0899)	(0.0942)	(0.0942)	(0.106)	(0.0881)	(0.0888)	
Party System Institutionalization	0.0405	0.0453	0.0383	0.0871	0.0837	0.0538	0.0369	
	(0.0910)	(0.0928)	(0.0926)	(0.106)	(0.110)	(0.0917)	(0.0925)	
Judicial Independence	0.267***	0.267***	0.275***	0.267***	0.279***	0.264***	0.265***	
	(0.0628)	(0.0632)	(0.0637)	(0.0642)	(0.0668)	(0.0628)	(0.0637)	
Civil Service Capacity	0.183***	0.180***	0.180***	0.154**	0.145**	0.196***	0.183***	
	(0.0570)	(0.0581)	(0.0576)	(0.0603)	(0.0628)	(0.0583)	(0.0574)	
Cabinet Stability	0.0423	0.0362	0.0195	-0.0512	-0.0509	0.0423	0.0444	
	(0.132)	(0.134)	(0.133)	(0.147)	(0.150)	(0.132)	(0.133)	
Political Regime		0.0132			0.00194			
		(0.0419)	0.00000		(0.0445)			
Proportional Electoral System			0.00260		-0.0493			
Effective New York Deltification			(0.0712)	0.0040	(0.0824)			
Effective number of Political Parties				0.0243	0.0258			
				(0.0161)	(0.0166)	0.0157		
Volatility of GDP						0.0157		
Volgtility of Torms of Trado						(0.0143)	0 1 9 2	
volatility of Terms of Trade							-0.165	
Observations	00	00	00	70	77	00	(0.000)	
R-squared	90 0 897	90 0 897	0.808	0 0 0 0 0 0	0.906	0 800	90 0 897	

 Table 4. Institutional Strengths and the Characteristics of Policies,

 Multivariate Regression

Standard errors in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Controls included in the regressions are: Ln(GDPpc) in 1990 and 7 regional dummies

The results are encouraging regarding the framework. In particular, the institutional variables related to intertemporal cooperation are significant in explaining the quality of policies, but the institutional variables traditionally used in the literature to explain economic outcomes seem not to matter that much, at least not directly. Adding variables that proxy the existence of "economic shocks," such as the volatility of GDP and the volatility of terms of trade, does not alter the results significantly.

## 6. Conclusions

The paper has introduced a number of cross-national measures of various properties or qualities of public policies. We have shown that certain characteristics of the policymaking environment, which we have related to the capacity of generating intertemporal agreements, tend to generate better policy features. For example, we show that countries with higher

<sup>&</sup>lt;sup>29</sup> We use the average from 1990 to 2004.

congress capabilities, judicial independence, and bureaucratic quality tend to produce higherquality policies.

While we believe these preliminary results are interesting in their own right, if for no other reason than to foster discussion on these subjects, we hope that the dataset itself will stimulate scholars to utilize it in their analysis of political institutions, public policies, and development outcomes.

The measures of policy characteristics that we have assembled can have uses beyond the one developed in this paper. We are pursuing two alternative uses of these variables. First, we use these variables to challenge some of the results of the veto player literature. One of the most prominent theoretical strands in comparative politics today is the veto player theory developed and summarized by Tsebelis (2002)-and applied to presidential democracies in Cox and McCubbins (2001). It is an approach that attempts to provide a synthetic characterization of political systems in terms of their impact on the ease or difficulty of implementing policy change, designating as veto players those political actors whose agreement is necessary to change policy. One of the main predictions of veto player theory is that polities with a higher number of veto players are less likely to change their policies. (This is good for sustaining policy commitments, but bad for adapting to changing circumstances or to policy failures.) Using the policy variables constructed in this paper, and the framework summarized in the previous section, Scartascini, Stein and Tommasi (2008) challenges this prediction from the veto player approach. We postulate that polities more able to sustain policies over time will not necessarily will be less able to adjust policies when necessary, and our separate notions of stability and adaptability attempt to capture these two distinct concepts. Furthermore, in our perspective, polities that are better able to cooperate over time might be able to achieve more of both desirable policy qualities in such a way that we could find these two variables positively correlated in a cross section of countries (as we indeed do). Furthermore, in that paper we find that a higher number of veto players indeed increases both stability and adaptability, and that both variables are better explained by institutional variables attempting to capture intertemporal cooperation than by variables that measure the number of veto players.

Second, we are also attempting to use these proxies for state capacity as control variables for explaining the impact of public spending on a number of social areas.

Preliminary results suggest that the policy environment does indeed matter. However, it also suggests that it matters differently depending on the issue at hand (in our example, education and health). We find that if a country's policy environment is bad, spending more on health has no clear effect on improving life expectancy. Conversely, as countries develop a good policy environment, they tend to benefit more from a given amount spent (the effect of health expenditures is positive and significant). With respect to education, both the general policy index and the public-regardedness of policies matter significantly, but countries displaying low-quality policies tend to benefit more from higher levels of spending. Those simple exercises suggest that the policy environment does indeed matter. However, they also suggest that the policy environment matters differently depending on the issue at hand. Despite the differences, a common message comes out: countries may achieve important gains if they could improve their institutions and the quality of policies they produce (as argued by other scholars, such as Killick, 1995). In some cases, such improvement may be enough to compensate for low levels of resources. In other cases, it might potentiate whatever is invested.

#### **Appendix:** A Simple Model of Political Cooperation and Public Policy

Imagine a game between two players or "groups," i = A, B. Each player tries to minimize:

$$\sum_{t=0}^{\infty} \delta^t E[L_i(y_t, \theta_t)]$$

where  $\delta \in [0,1]$  is a (common) discount factor measuring "patience," and  $L_i()$  is a loss function that depends on the "collectively" chosen policy y and the economic shock  $\theta$ , identically and independently distributed over time, with  $E(\theta) = 0$ . For simplicity, let:

$$L_i(y_t, \theta_t) = [y_t - (y_i + \theta_t)]^2$$

The fact that  $y_A \neq y_B$  captures the elements of conflict, while the fact that everybody's preferred policy responds in the same direction to the economic shock  $\theta$ captures the common interest, or economic efficiency. Assume that  $y_B = -y_A > 0$ .

In each period, after the random shock  $\theta_t$  is realized, the policy  $y_t$  is decided through a collective choice mechanism. Also assume that the recognition rule,  $\mu_t = i$ , generates an equal probability that each player  $i \in \{A, B\}$  be the one-period dictator ( $\mu_t = i$  implies that player *i* decides  $y_t$  in period *t*.) That is:<sup>30</sup>

$$\mu_t = i, i \in \{A, B\}$$
 with prob. 0.5

Assume furthermore that there is an initial period (zero) in which, by unanimity, players can make some agreements.

We start defining a first-best utilitarian benchmark as:

$$\operatorname{Min} \sum_{t=0}^{\infty} \delta^{t} E \big[ L_{A}(y_{t}, \theta_{t}) + L_{B}(y_{t}, \theta_{t}) \big]$$

Given our assumptions, the above minimization simplifies to:

$$\operatorname{Min} \sum_{t=0}^{\infty} \delta^{t} E(y_{t} - \theta_{t})^{2}$$

<sup>&</sup>lt;sup>30</sup> This is a very simplified version of several richer collective decision-making mechanisms, such as those in Alesina (1988), Baron and Ferejohn (1989) and Dixit, Grossman and Gul (2000).

so that the first-best policy is  $y_t = y^*(\theta_t) = \theta_t$  for all *t*. This result indicates that the firstbest policy is a function of the realization of economic shocks, but independent of the realization of political shocks.

We analyze now the solution to the non-cooperative game. The one-shot Nash equilibrium has  $y_t = y_{\mu_t} + \theta_t$ . That is, each political player implements his or her most desired policy, ignoring the interest of others. Turning to the repeated game, the infinite repetition of one-shot Nash is always an equilibrium. We define  $V^N$  as the present value of expected loss for each player from the infinite repetition of the one-shot Nash equilibrium. Then we have that:

$$V^{N} = \frac{1}{2} \left[ 0 + (2y_{B})^{2} \right] + \delta V^{N} = \frac{2y_{B}^{2}}{1 - \delta}$$

To simplify the analysis we focus now on the possibility of the most cooperative behavior being supported by the punishment strategy of permanent reversion to non-cooperation (as in Dixit et al., 2000, and Dixit, 2003). This is strategy (S1) for both i:

$$y_1^{\iota} = y^{\star}(\theta_1) = \theta_1$$

and

$$y_t^i = \begin{cases} y_t^* = \theta_t & if \quad y_\tau = \theta_\tau \quad \forall \, \tau < t \\ y_i + \theta_t & otherwise \end{cases}$$

The payoff along the equilibrium path of cooperation is:

$$V^* = \frac{y_B^2}{1 - \delta}$$

for both players, which coincides with the value of the loss function in the first best scenario. In order to verify whether this strategy pair constitutes an equilibrium, we have to consider the value of an opportunistic deviation to  $y_{\mu_t} + \theta_t$ . Such deviation would move the game to non-cooperation forever, leading to the value:

$$V^{D} = 0 + \delta V^{N} = \frac{\delta}{1 - \delta} 2y_{B}^{2}$$

Comparing the loss functions  $V^*$  and  $V^D$ , we can conclude that cooperation can be sustained if  $\delta \ge 1/2$ . The first-best can be attained, then, for  $\delta$  large enough, that is, when players have long horizons.

Consider now what might transpire in the previous stage of the game, what we call the contracting moment. The features of the resulting policies will depend on the set of feasible contracts. Assume that agreements can be enforced, but that the realization of economic shocks is not verifiable. In that case, it will not be possible to sign contracts that prescribe (economic) state – contingent contracts. On the other hand, the parties can agree on simple "rules." In our example, it can be shown that the best such rule is to set  $y_t = 0$  for all t. This will deliver an expected loss of:

$$V^{0} = \sum_{t=0}^{\infty} \delta^{t} E(y_{B} + \theta_{t})^{2} = \frac{y_{B}^{2} + Var(\theta)}{1 - \delta}$$

This outcome is inferior to the first best for both players, i.e.,  $V^0 > V^*$ . (Remember that we are dealing with *loss* functions.) The discretionary cooperative equilibrium is preferable to a rigid rule. Thus, whenever the repeated game delivers cooperation, a rigid rule will not be utilized. Comparing the rigid rule to the non-cooperative case, we have that  $V^N > V^0$  if  $(y_B)^2 > Var(\theta)$ . This implies that when the parties have a limited capacity to self-enforce cooperative agreements (i.e., when  $\delta$  is low), rigid policy rules (not responsive to the economic environment) will be chosen if the conflict of interest  $(y_B)$  is large compared to the volatility of the economic environment  $(Var(\theta))$ . Thus, we find that when there is a low capacity to enforce intertemporal political exchanges, depending on the extent of the distributive conflict as related to the nature of economic volatility, we may observe highly volatile political agreements or highly inflexible policies.

## References

- Alesina, A. 1988. "Credibility and Policy Convergence in a Two-Party System with Rational Voters." *American Economic Review* 78(4): 796-805.
- Baron, D.P., and J.A. Ferejohn. 1989. "Bargaining in Legislatures." *American Political Science\_Review* 83: 1181-1206.
- Barro, R.J., and D.B. Gordon. 1983. "A Positive Theory of Monetary Policy in a Natural-Rate Model." *Journal of Political Economy* 91(4): 589-610.
- Berkman, H. et al. 2009. "Political Institutions, State Capabilities, and Public Policy: An International Dataset." Washington, DC, United States: Inter-American Development Bank. Available at:

http://www.iadb.org/RES/pub\_List.cfm?pub\_topic\_id=DBA&type=pub\_type&pub\_ty pe\_id=DBA&pub\_type\_id1=DBA&language=english

- Besley, T., and T. Persson. 2007. "The Origins of State Capacity: Property Rights, Taxation, and Politics." NBER Working Paper 13028. Cambridge, United States: National Bureau of Economic Research.
- Braun, M., and L. Di Gresia. 2002. "Towards Effective Social Insurance in Latin America: The Importance of Countercyclical Fiscal Policy." Research Department Working Paper 487. Washington, DC, United States: Inter-American Development Bank.
- Calvo, G.A. 1996. Money, Exchange Rates, and Output. Cambridge, United States: MIT Press.
- Calvo, G.A., and A. Drazen. 1998. "Uncertain Duration of Reform: Dynamic Implications." *Macroeconomic Dynamics* 2(4): 443-55.
- Cox, G.W., and M.D. McCubbins. 2001. "The Institutional Determinants of Economic Policy Outcomes." In: S. Haggard and M.D. McCubbins, editors. *Presidents, Parliaments* and Policy. New York, United States: Cambridge University Press.
- Crain, W.M. 2001. "Institutions, Durability and the Value of Political Transactions." In. W.F.Shugart II and L. Razzolini, editors. *The Elgar Companion to Public Choice*.Cheltenham, United Kingdom: Edward Elgar.
- Crain, M., and R. Tollison. 1979. "Constitutional Change in the Interest-Group Theory of the Government." *Journal of Legal Studies* 8: 165-75.

- De Figueiredo, R.J.P. 2002. "Electoral Competition, Political Uncertainty, and Policy Insulation." *American Political Science Review* 96(2): 321-333.
- Diaz, R., et al. 2009. "Assessing the Quality of Public Policy" A New Cross-country Dataset." London, United Kingdom: London School of Economics. Mimeographed document.
- Dixit, A.K. 1996. *The Making of Economic Policy: A Transaction-Cost Politics Perspective*. Cambridge, United States: MIT Press.
- ----. 2003. "Some Lessons of Transaction Cost Politics for Less-Developed Countries." *Economics and Politics* 15(2): 107-133
- Dixit, A.K., G. Grossman and F. Gul. 2000. "The Dynamics of Political Compromise." *Journal of Political Economy* 108(3): 531-68.
- Dominguez, J., and A. Lowenthal. 1996, *Constructing Democratic Governance: South America in the 1990s*. Baltimore, United States: Johns Hopkins University Press.
- Drazen, A. 2000. *Political Economy in Macroeconomics*. Princeton, United States: Princeton University Press.
- Engerman, S.L., and K.L. Sokoloff. 2008. "Debating the Role of Institutions in Political and Economic Development: Theory, History, and Findings." *Annual Review of Political Science* 11: 119-135.
- Epstein, L., and S. O'Halloran. 1999. *Delegating Powers: A Transaction Cost Politics Approach to Policy Making Under Separate Powers*. New York, United States: Cambridge University Press.
- Fudenberg, D., and E. Maskin. 1986. "The Folk-Theorem in Repeated Games with Discounting and with Incomplete Information." *Econometrica* 54: 533–56.
- Fudenberg, D., and J. Tirole. 1991. Game Theory. Cambridge, United States: MIT Press.
- Gerring, J., S. Thacker and C. Moreno. 2005. "Centripetal Democratic Governance: A Theory and Global Inquiry." *American Political Science Review* 99(4): 567-581.
- Green, E., and R. Porter. 1984. "Non-cooperative Collusion under Imperfect Price Information." *Econometrica* 54: 975–94.
- Henisz, W.J. 2002. "The Institutional Environment for Infrastructure Investment." *Industrial and\_Corporate Change* 11(2): 355-389.
- Huber, J., and C. Shipan. 2002. *Deliberate Discretion? The Institutional Foundations of Bureaucratic Autonomy*. New York, United States: Cambridge University Press.

- Inter-American Development Bank (IDB). 2005. *The Politics of Policies*. Economic and Social Progress Report 2006. Washington, DC, United States: IDB.
- Jones, M. 2009. "The Role of Political Parties and Party Systems in the Policymaking Process." Forthcoming in: C. Scartascini, E. Stein and M. Tommasi, editors. *Political Institutions, Actors and Arenas in Latin American Policymaking.* London, United Kingdom: Palgrave.
- Killick, T. 1995. The Flexible Economy: Causes and Consequences of the Adaptability of National Economies. London, United Kingdom: Routledge.
- Knack, S., and N. Manning, 2000, Towards Consensus on Governance Indicators: Selecting Public Management and Broader Governance Indicators. Washington, DC, United States: World Bank.
- Landes, W., and R. Posner. 1975. "The Independent Judiciary in an Interest Group Perspective." *Journal of Law and Economics* 18: 875-901.
- Levy, B., and P.T. Spiller. 1994. "The Institutional Foundations of Regulatory Commitment: A Comparative Analysis of Telecommunications Regulation." *Journal of Law, Economics\_and Organization* 10(2): 201-246.
- Machado, F. et al. 2009. "Political Institutions and Policy Outcomes around the World; How Intertemporal Transactions Matter for the Quality of Public Policies." Washington, DC, United States: Inter-American Development Bank. Mimeographed document.
- Mailath, G. and L. Samuelson, 2006. *Repeated Games and Reputations. Long-Run Relationships*. Oxford University Press.
- Mainwaring, S., and T. Scully. 1995. *Building Democratic Institutions: Party Systems in Latin America*. Stanford, United States: Stanford University Press.
- Mainwaring, S., and M. Torcal. 2005. "Party System Institutionalization and Party System Theory: After the Third Wave of Democratization." Paper presented at the annual meeting of the American Political Science Association, Washington, DC, United States, September.
- Martínez-Gallardo, C. 2009. "Cabinets and the Policy-Making Process." Forthcoming in: C. Scartascini, E. Stein and M. Tommasi, editors. *Political Institutions, Actors and Arenas in Latin American Policymaking*. London, United Kingdom: Palgrave.

- Moe, T. 1990. "Political Institutions: The Neglected Side of the Story." *Journal of Law, Economics and Organization* 6: 213-253.
- North, D.C. 1990. "A Transaction-Cost Theory of Politics." *Journal of Theoretical Politics* 2(4): 355-67.
- Persson, T., and G. Tabellini. 2000. *Political Economics*. Cambridge, United States: MIT Press.
- ----. 2003. The Economic Effects of Constitutions. Cambridge, United States: MIT Press.
- Przeworski, A. 2005. "Democracy as an Equilibrium." Public Choice 123: 253-273.
- Profils Institutionnels. Various dates. Centre D'Estudes Prospectives et D'Informations Internationales. Available at: http://www.cepii.fr/ProfilsInstitutionnelsDatabase.htm
- Rodrik, D. 1989. "Policy Uncertainty and Private Investment in Developing Countries." NBER Working Paper 2999. Cambridge, United States: National Bureau of Economic Research.
- ----. 1995. "Political Economy of Trade Policy." In: G.M. Grossman and K. Rogoff, editors. *Handbook of International Economics*. Volume 3. Amsterdam, The Netherlands: Elsevier.
- Rotemberg, J., and G. Saloner. 1986. "A Supergame-Theoretic Model of Price Wars during Booms." *American Economic Review* 76: 390–407.
- Saiegh, S. 2009. "Active Players or Rubber Stamps? An Evaluation of the Policy-Making Role of Latin American Legislatures." Forthcoming in: C. Scartascini, E. Stein and M. Tommasi, editors. *Political Institutions, Actors and Arenas in Latin American Policymaking*. London, United Kingdom: Palgrave.
- Scartascini, C. 2007. "The Institutional Determinants of Political Transactions." Research Department Working Paper 580. Washington, DC, United States: Inter-American Development Bank.
- Scartascini, C., E. Stein and M. Tommasi. 2008a. "Veto Players, Intertemporal Interactions and Policy Adaptability: How Do Political Institutions Work?" Research Department Working Paper 645. Washington, DC, United States: Inter-American Development Bank.

- ----. 2008b. "Political Institutions, State Capabilities and Public Policy: International Evidence." Research Department Working Paper 661. Washington, DC, United States: Inter-American Development Bank.
- Shepsle, K., and M. Bonchek. 1997. *Analyzing Politics: Rationality, Behavior and Institutions*. New York, United States: Norton.
- Shepsle, K., and B. Weingast. 1995. *Positive Theories of Congressional Institutions*. Ann Arbor, United States: University of Michigan Press.
- Souza, M. 2009. "Judicial Roles in the Policy-Making Process and Public Policy in Latin America." Forthcoming in: C. Scartascini, E. Stein and M. Tommasi, editors. *Political Institutions, Actors and Arenas in Latin American Policymaking*. London, United Kingdom: Palgrave.
- Spiller, P.T., E. Stein and M. Tommasi. 2003. Political Institutions, Policymaking Processes, and Policy Outcomes: An Intertemporal Transactions Framework. Washington DC, United States: Inter-American Development Bank. Mimeographed document.
- Spiller, P.T., and M. Tommasi. 2007. *The Institutional Foundations of Public Policy in Argentina*. Cambridge, United Kingdom: Cambridge University Press.
- Stein, E., and M. Tommasi. 2007. "The Institutional Determinants of State Capabilities in Latin America." In: F. Bourguignon and B. Pleskovic, editors. Annual World Bank Conference on Regional Development Economics: Beyond Transition. Washington, DC, United States: World Bank.
- Stein, E. et al. 2008. Policymaking in Latin America: How Politics Shapes Policies. Cambridge, United States: Harvard University, David Rockefeller Center for Latin American Studies.
- Tsebelis, G. 2002. *Veto Players: How Political Institutions Work*. Princeton, United States: Princeton University Press.
- Van de Walle, S. 2005. "Measuring Bureaucratic Quality in Governance Indicators." Paper prepared for the 8th Public Management Research Conference, Los Angeles, United States.
- Wantchekon, L. 2000. "Credible Power-Sharing Agreements." Constitutional Political Economy 11(4): 329-352.

- Weaver, R.K., and B.A. Rockman. 1993. *Do Institutions Matter? Government Capabilities in the\_United States and Abroad.* Washington, United States: Brookings Institution Press.
- Weingast, B., and W. Marshall. 1988. "The Industrial Organization of Congress; or, Why Legislatures, like Firms, Are Not Organized as Markets?" *Journal of Political Economy* 96(11): 132-163.
- Zuvanic, L., and M. Iacoviello. 2009. "The Bureaucracy in Latin America." Forthcoming in:C. Scartascini, E. Stein and M. Tommasi, editors. *Political Institutions, Actors and Arenas in Latin American Policymaking*. London, United Kingdom: Palgrave.